Author's response to reviews

Title: Intraoperative cerebral oxygen saturation trend of values and its relationship with cognitive decline after total knee replacement

Authors:

Fátima Salazar (12053fsg@comb.cat)
Marta Doñate (mrtdonate@hotmail.com)
Teresa Boget (tboget@clinic.ub.es)
Ana Bogdanovich (abogdano@clinic.ub.es)
Misericordia Basora (mbasora@clinic.ub.es)
Ferran Torres (ferran.torres@uab.es)
Isabel Gracia (igracia@clinic.ub.es)
Neus Fàbregas (fabregas@clinic.ub.es)

Version: 2
Date: 3 December 2013

Author's response to reviews: see over
Name of corresponding author: Fátima Salazar García

Title of the manuscript: “Intraoperative cerebral saturation trend of values and its relationship with cognitive decline after knee replacement”

We are submitting the enclosed material for possible publication in BMD Anesthesiology. It is not being considered for publication nor has been published in whole or in part elsewhere.

All authors have contributed intellectually to the work. Participate in the work and read the manuscript before its submission for publication.

The authors declare that they have no competing interests.

I have made de changes following your indications:

- This study has been possible thank you to a Spanish National Grant: FIS PI050066 (principal investigator Dra. Fatima Salazar). We have not had any other financial support. The design, explorations, statistic analysis, etc., were covered by the National Grant, including the translation of the manuscript by a language editor.

- Following your indications, we have include an acknowledgment to our language editor. We thank Mary Ellen Kerans who provided medical writing services, being supported by the aforementioned National Grant.